Author's response to reviews

Title: Can pill placebo augment cognitive-behavior therapy for panic disorder?

Authors:

Toshi A Furukawa (furukawa@med.nagoya-cu.ac.jp)
Norio Watanabe (watanori@yha.att.ne.jp)
Ichiro M Omori (mmds@sannet.ne.jp)
Rachel Churchill (Rachel.Churchill@iop.kcl.ac.uk)

Version: 4 Date: 12 December 2007

Author's response to reviews:

Formatting changes requested
-----------------------------
Please include a Conclusions section between the Discussions and Competing interests sections. --> DONE.

Acknowledgements --> I HAVE NONE.

EndNote or Refman references - There is a discrepancy between the number of references in your manuscript, and the number extracted by our system, for example References 14 and 15 from the references list in the manuscript are missing in the extracted references list. We have a helpguide http://www.biomedcentral.com/info/ifora/endnote/#technicalhelp, which might be able to help you improve this.

--> I REFORMATTED THE MANUSCRIPT AND IT SHOULD NOW BE OK.

Table- please rename this Table 1 --> DONE.

Figure cropping --> DONE. THIS IS THE SMALLEST ONE CAN GET.